Author's response to reviews

Title: Effectiveness of a transitional home care program in reducing acute hospital utilization: A retrospective study

Authors:

Lian Leng Low (low.lian.leng@sgh.com.sg)
Farhad Fakhrudin Vasanwala (farhad.fakhrudin.vasanwala@sgh.com.sg)
Lee Beng Ng (ng.lee.beng@sgh.com.sg)
Cynthia Chen (a0024354@nus.edu.sg)
Kheng Hock Lee (lee.kheng.hock@sgh.com.sg)
Shu Yun Tan (tan.shu.yun@sgh.com.sg)

Version: 2 Date: 5 August 2014

Author's response to reviews: see over
Cover Letter detailing changes

1. Added in the title page at the front of the main manuscript body
2. Added in abstract with keywords to the main manuscript
3. Added in page numbers and line numbers
4. Added in list of abbreviations after conclusion
5. Added in a section on Endnotes after acknowledgements